Author’s response to reviews

Title: Transferring Research from a University to the UK National Health Service: The Implications for Impact

Authors:

helen payne (h.l.payne@herts.ac.uk)

Version: 2 Date: 24 May 2017

Author’s response to reviews:

the references in text and list have been re numbered sequentially as requested.